



Article Does the Digital Economy Promote the Reduction of Urban Carbon Emission Intensity?

Shouwu Jing¹, Feijie Wu¹, Enyi Shi¹, Xinhui Wu¹ and Minzhe Du^{2,*}

- ¹ School of International Trade, Shanxi University of Finance and Economics, Taiyuan 030006, China
- ² School of Economics and Management, South China Normal University, Guangzhou 510006, China
- * Correspondence: mzdu@m.scnu.edu.cn

Abstract: The impact of the digital economy is increasing, and its environmental effect has attracted more and more attention. The digital economy promotes the improvement of production efficiency and the government's environmental governance capacity, and contributes to the reduction of urban carbon emission intensity. In order to study the impact of digital economy development on urban carbon emission intensity, this paper analyzes the theoretical basis of the digital economy on the reduction of carbon emission intensity, and then, based on the panel data of cities from 2011 to 2019, uses the two-way fixed effect model for empirical testing. The regression results show that the development of the digital economy has promoted the reduction of carbon emission intensity of cities, promoted the green transformation and upgrading of cities, and lays a foundation for China to achieve carbon peaking and carbon neutralization through the improvement of human capital investment and green innovation level. The basic conclusion is robust by changing core explanatory variables, changing samples, replacing regression methods, and shrinking and truncating tests. The impact of the digital economy on urban carbon emission intensity varies with the location, grade and size of the city. Specifically, the development of the digital economy in cities in the eastern and central regions, cities at or above the sub provincial level, large cities and non-resource-based cities has promoted the reduction of urban carbon emission intensity. In terms of resource-based cities, the development of the digital economy in renewable resource-based cities and resource-based cities dominated by iron ore and oil mining has promoted the decline in urban carbon emission reduction intensity.

Keywords: digital economy; carbon emission intensity; human capital investment; green innovation; two-way fixed effect model

1. Introduction

The digital economy is a digital transformation in the process of economic development based on digital technologies, including 5G, artificial intelligence, big data, etc. The digital economy has a bearing on the overall development of a country, becoming an important field for countries around the world to seize the commanding heights of economic and social development, and has a profound impact on global development [1–3]. In 2020, the scale of China's digital economy was 39.2 trillion yuan, accounting for 38.60% of GDP. From 2016 to 2020, the total volume of China's digital economy increased by 1.74 times, with an annual average growth rate of 17.08%. The growth rate of the digital economy is much higher than that of GDP in the same period. As far as the development of urban digital economy, a development pattern has taken shape overall, with "Beijing, Shanghai, Guangzhou, Shenzhen and Hangzhou" as the lead and provincial capitals as the main part. Beijing, Shanghai and Hangzhou are the top three cities in the overall ranking of the digital economy index. Cities in the eastern region perform better in terms of digital economy policy and environment, scale and quality of the digital economy. Guiyang performed better in digital economy policy and environment, while Chengdu, Chongqing and Xi'an



Citation: Jing, S.; Wu, F.; Shi, E.; Wu, X.; Du, M. Does the Digital Economy Promote the Reduction of Urban Carbon Emission Intensity? *Int. J. Environ. Res. Public Health* **2023**, 20, 3680. https://doi.org/10.3390/ ijerph20043680

Academic Editors: Jiaxing Cui, Jing Luo, Ying Jing and Liqun Sun

Received: 16 January 2023 Revised: 16 February 2023 Accepted: 16 February 2023 Published: 19 February 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). ranked better in digital economy scale and quality. The development of China's digital economy has made great progress. The digital economy can effectively support sustainable economic development [4], which has become a new driving force for China's high-quality economic development [5,6]. Green development is an important part of high-quality economic development., and achieving carbon peak and carbon neutralization is also an important starting point for green economic development. Governments around the world have taken many steps to reduce CO₂ emissions [7–11].

In terms of carbon emissions, China's carbon emissions increased from 8145.8 million tons in 2010 to 9899.3 million tons in 2020, with an average annual growth rate of 2.4%. In 2021, China's carbon emissions accounted for 31.116% of the world's total carbon emissions [12]. China is the largest emitter of carbon dioxide in the world [13]. The rapid growth of "volume" and "speed" of carbon emissions makes the Chinese government face huge pressure to reduce emissions in the international community. However, the Chinese government attaches great importance to the climate and ecological environment changes caused by carbon emissions, and China proposed to achieve carbon peaking by 2030 and carbon neutrality by 2060 at the General debate of the 75th session of the UN General Assembly in 2020. All countries in the world regard low-carbon green development as an important means to maintain competitive advantage. Low carbon development is not only a key link in China's transformation and upgrading, but also an important focus for China to participate in world competition.

The digital economy has significantly changed the production and organization form. We would wonder whether the development of the digital economy could have a positive impact on the green development of the economy? Can the development of the digital economy effectively promote the reduction of carbon emission intensity? If the development of the digital economy effectively reduces the intensity of carbon emissions, what are the mechanisms behind it? Could the effect of reducing carbon emission intensity of the digital economy have different results depending on the region, size, level and resource dependence of the city? Although many scholars have studied the economic and social development effects brought by the digital economy and the influencing factors of carbon emission, respectively, there is a lack of detailed research on the carbon emission intensity of the digital economy development on China's carbon emission intensity and the direction of China's carbon emission reduction efforts, and provide an important reference for China's economic development and environmental protection.

In the face of the increasingly serious energy crisis, global climate deterioration, ecological environment damage, and the pressure of China's economic transformation and upgrading, it is of great significance to study the impact of the digital economy on urban carbon emission intensity, whether the development of the digital economy can achieve ecological environment protection. Research has shown that the digital economy can effectively promote economic transformation and upgrading [14], but can the digital economy reduce carbon emission intensity, and how? There is no clear answer. Therefore, this paper analyzes the theoretical mechanism of the digital economy to reduce carbon emission intensity, and based on the panel data of cities from 2011 to 2019, uses the two-way fixed effect model to empirically test the impact, mechanism and heterogeneity of digital economy development on urban carbon emission intensity. The research results show that first, the development of the digital economy has significantly reduced the intensity of urban carbon emissions. Second, the digital economy reduces the intensity of urban carbon emissions by increasing the level of human capital investment and improving the level of green innovation. Third, the digital economy has a heterogeneous effect on the reduction of urban carbon emission intensity. Specifically, the development of the digital economy has a significant effect on the reduction of carbon emission intensity of cities in the eastern and central regions, cities above the sub provincial level, large cities and nonresource-based cities. For resource-based cities, it is mainly renewable resource-based cities

and resource-based cities dominated by iron ore and oil that have achieved a reduction in carbon emission intensity.

The possible marginal contributions of this paper are as follows. First, most of the existing research literatures do not link the development of the digital economy with carbon emissions. More attention is focused on the scale measurement of the digital economy and its impact on economic transformation and upgrading, and less on the impact of the development of the digital economy on the intensity of carbon emissions. Therefore, this paper makes up for this shortcoming and analyzes the impact of the development of the digital economy on carbon emission intensity from the urban level. Second, most of the existing literatures focus on the research on the measurement of digital economy development level and carbon emissions at the national and provincial levels. This paper analyzes from a more microscopic perspective, that is, the urban level, and further finds the difference of the effect of the digital economy on the reduction of carbon emission intensity in different heterogeneous cities. Third, this paper examines the mechanism of reducing urban carbon emission intensity by the digital economy. Specifically, the digital economy can reduce the intensity of urban carbon emissions by improving the level of human capital investment and the level of green innovation, and can be more targeted and flexible in policy formulation through the identification effect mechanism.

The rest of this paper is arranged as follows. Section 2 contains the literature review and theoretical basis. Section 3 contains the variables and model selection. Section 4 contains the basic regression analysis, including basic regression results, robustness test and endogenous test. Section 5 contains the mechanism test. Section 6 contains further analysis. Section 7 contains the conclusions and policy recommendations.

2. Literature Review and Theoretical Basis

2.1. Literature Review

The literature review part analyzes the impact of the digital economy on economic development and the factors affecting carbon emissions.

2.1.1. The Impact of Digital Economy on Economic Development

The role of the digital economy in economic and social development is becoming more and more obvious. Countries around the world have taken improving the development level of the digital economy as the commanding height of future world competition. The digital economy can effectively promote economic development [14]. It can not only expand the economic scale, but also promote high-quality economic development.

From a macro perspective, the digital economy promotes high-quality economic development by improving the quality of research and development, upgrading the industrial structure and improving total factor productivity. The digital economy can effectively improve the quality and efficiency of R&D and promote high-quality economic development. On the one hand, it can reduce the restrictive impact of time and space on the interaction of R&D personnel, on the other hand, it can promote the shortening of R&D cycle and improve the development efficiency of new products and technologies [15]. Additionally, the digital economy can realize the upgrading effect of industrial structure and promote high-quality economic development. It can effectively promote the coordination between product production departments and realize digital manufacturing. Wallace [16] believes that the digital economy could unleash the potential of economic growth and promote inclusive economic development. Moreover, the digital economy improves total factor productivity and promotes high-quality economic development. The improvement of total factor productivity is the key to the long-term stable development of China's economy and the successful transformation of China's economy [17,18]. It can improve total factor productivity [19,20], but there are regional differences in its role. As the research of Yang and Lu [21] shows, the productivity level is low and the total factor productivity in the central and western regions has more room to improve. Therefore, the development of the

digital economy plays a more significant role in improving the productivity level and the total factor productivity in the central and western regions.

From the micro perspective, the digital economy can improve the decision-making level of enterprise managers, improve the input-output efficiency and improve the level of entrepreneurship, and promote high-quality economic development. The digital economy promotes the improvement of enterprise management, improves organizational performance [22], and thus improves the level of high-quality economic development. Jiao [23], taking JingDong as an example, found that the digital economy has promoted the reorganization of JingDong group's capabilities in terms of opportunity grasping, structural change, etc., and has embedded the most important data elements in the digital economy into the entire process of enterprise operation, achieving significant improvement in digital capabilities. Therefore, the digital economy can effectively promote the growth of enterprises, and further promote the long-term economic growth and structural upgrading. Meanwhile, the development of the digital economy can also effectively improve the inputoutput efficiency of enterprises and improve the quality of economic development. The digital transformation of enterprises not only changes the original organizational structure and governance structure of enterprises [24], but also improves the communication efficiency of enterprises, reduces resource redundancy, makes it more in line with the needs of marketization, and further improves the input and output efficiency of enterprises [25]. Moreover, the digital economy has further improved the quality of economic development by promoting the quality of entrepreneurship. The development of the digital economy has promoted the development of entrepreneurship ecosystem [26], which is more conducive to entrepreneurship [27]. On the one hand, it has accelerated the withdrawal of low-quality entrepreneurship from the market, on the other hand, it has promoted the entrepreneurship of general industry workers [28]. For rural residents, the digital economy has also improved the entrepreneurial environment and increased entrepreneurial opportunities, provided rural areas with opportunities to take advantage of digital technologies [29], activated entrepreneurial activities of rural residents [30], driven economic development in rural areas, and narrowed the urban-rural gap.

However, while the digital economy plays a positive role, it may also have negative effects. Syed et al. [31] takes digital finance as an example and believes that digital finance may also produce systemic risks and lead to the reduction of financial stability. Rochet and Tirole [32] believe that the digital economy may cause the contraction of traditional economy, reducing economic benefits.

2.1.2. Influential Factors of Carbon Emissions

Global warming affects economic and social development [33], so low-carbon economic development is crucial. The existing literature has explored the factors affecting carbon emissions, and the impact of macro carbon emission reduction policies on carbon emissions. Administrative carbon emission reduction policies mainly include environmental regulations and environmental goal constraints [34,35]. Administrative carbon emission reduction policies can promote the reduction of carbon emissions and carbon emission intensity. However, at the same time, it may also cause a distortion in resource allocation and a decline in welfare. Strict environmental regulations may change producers' expectations, increase carbon emissions in the short term [36], and even lead to pollution migration [37]. Furthermore, the impact of market mechanism on carbon emissions is noteworthy; it is generally believed that market mechanism can effectively reduce resource distortion, reduce energy consumption intensity and reduce carbon emissions. Tang et al. [38] found that carbon trading in pilot areas significantly promoted the reduction of carbon emission level. Higher levels of human capital and technology transfer from developed countries can contribute to the decline in CO₂ emissions [39].

2.2. Theoretical Analysis and Research Hypothesis

The digital economy is the fourth economic form after agricultural economy, industrial economy and information economy. It is a new economic form developed mainly by using big data, artificial intelligence, the Internet of Things, 5G and others [15,40]. First, the development of the digital economy has promoted the improvement of production efficiency, which not only improves the labor production efficiency, but also improves the green total factor productivity, thereby reducing the intensity of urban carbon emissions. The development of the digital economy has promoted the increase in urban output, especially the increase in high-quality output, and reduced the emission level of pollutants. Specifically, the digital economy promotes the coordinated development of various elements, realizes the full utilization of resources, improves the efficiency of resource allocation, improves the efficiency of energy resource utilization, reduces the resource redundancy of the main factor users, reduces the level of pollution emissions, and promotes the reduction of urban carbon emission intensity. At the same time, the digital economy has promoted the upgrading of urban industrial structure through the use of new technologies and new means, moving from the traditional industrial structure relying on high input, low output and high pollution emissions to high-end manufacturing, and realizing the coordinated development and mutual integration of manufacturing and modern service industries. The intensity of urban carbon emissions has been further reduced through the upgrading of industrial structure. In addition, with the in-depth development of the digital economy, data has become an important strategic asset. At the same time, China has accelerated the promotion of the national strategy of big data and included data as an emerging factor in the factor marketization reform. The development of the digital economy has greatly promoted the positive role of data as a factor of production in promoting enterprises' total factor productivity and even green total factor productivity. The development of the digital economy not only improves the information transparency of the use of data elements, but also improves the organizational and operational performance of enterprises through the use of big data. The development of the digital economy integrates land, labor, capital, technology and data in a coordinated manner, which increases the output of enterprises and reduces carbon emissions, reducing the intensity of carbon emissions.

Second, the development of the digital economy has improved the government's governance ability, especially the environmental governance ability, and promoted the reduction of urban carbon emission intensity. On the one hand, the digital economy has improved the level of government governance and reduced the intensity of urban carbon emissions. The improvement of the government's environmental governance level can not only effectively reduce haze pollution, but also significantly promote high-quality economic development. As far as carbon emissions are concerned, the construction of low carbon cities represents the governance level of cities in terms of low carbon. During the construction of low carbon cities, local governments have explored their own low-carbon development path, improved the environmental governance level of local governments, and achieved the effect of improving government air quality. On the other hand, the digital economy has also improved the government's supervision and monitoring capacity. The government's monitoring of pollution discharge and data collection have become more effective, forming a systematic monitoring system, standardized monitoring methods and timely monitoring feedback.

Hypothesis 1: *The development of the digital economy has promoted the reduction of urban carbon emission intensity and achieved green development.*

2.2.1. Improvement of Human Capital Investment

The digital economy has effectively reduced the intensity of urban carbon emissions by promoting urban human capital investment. Human capital investment includes not only the investment in compulsory education, high school and university, but also the investment in vocational ability training. The improvement of human capital investment can significantly improve the level of human capital, which can effectively promote highquality economic development. Li et al. [41] believed that the government should actively increase the investment in human capital, thus promoting the effective supply of human capital and achieving high-quality economic development.

On the one hand, the digital economy drives the improvement of human capital level by promoting the improvement of human capital investment, thus combining with artificial intelligence, big data, industrial robots and other technologies of the digital economy. High quality human capital can further contribute to the multiplier effect of human capital investment, and expand the positive role of the digital economy in environmental protection and environmental governance. On the other hand, green development or low-carbon development itself requires workers to reserve certain skills, knowledge and quality. If the level of human capital is low, it is difficult to adapt to the needs of green development, and the positive role of green development is also limited. The higher the level of human capital, the stronger the ability to have green jobs, and the more conducive to promoting green economic development. Therefore, the development of the digital economy has promoted the improvement of human capital investment, further accelerated workers' mastering of new knowledge and technology, promoted the accumulation and improvement of human capital, and thus matched with green development. In addition, with China's economic transformation and development, China's economy is increasingly demanding green development. The higher the level of human capital, the easier it is to find jobs that match green jobs.

Therefore, the higher the investment in human capital, the higher the level of human capital. The stronger the role of the digital economy in ecological environment protection, the more conducive it is to reduce the intensity of urban carbon emissions and improve the level of urban ecological environment.

Hypothesis 2: *The development of the digital economy further promotes the reduction of urban carbon emission intensity by increasing human capital investment.*

2.2.2. Improvement of Green Innovation Level

Technological progress is the driving force for overcoming the middle-income trap [42], and also the key to achieving green economic development. Green innovation plays an important role in achieving "green water and green mountains are golden mountains and silver mountains". Green innovation is the basis for achieving urban economic development and environmental protection, and helps to reduce urban carbon emission intensity. The digital economy is a new driving force to improve innovation level [43]. The digital economy can effectively reduce urban carbon emission intensity by promoting urban green innovation level.

The digital economy can improve the level of urban green innovation by improving the efficiency of resource allocation, reducing financing constraints and green knowledge spillovers, and thus reduce the intensity of urban carbon emissions. Firstly, the digital economy has improved the efficiency of resource allocation, which makes it possible to further enhance the level of green innovation. Compared with traditional innovation activities, green innovation faces greater uncertainty and requires higher financial support. The digital economy can improve the efficiency of resource allocation and improve the production efficiency of enterprises. It not only saves production costs, but also improves the retained earnings of enterprises, providing internal financial support for enterprises to actively carry out green innovation. Secondly, the development of the digital economy promotes the reduction of urban carbon emission intensity by reducing the constraint level of green innovation financing. The development of the digital economy can improve the probability of information collection of banks and other institutions, reduce the degree of information asymmetry between banks and enterprises, reduce the cost of information communication, improve the matching quality between banks and enterprises and other economic entities, thus further reducing the external financing constraints faced by enterprises in green innovation. The decline of external financing constraints of enterprises has promoted the improvement of green innovation level of enterprises. The improvement of

urban green innovation level can effectively reduce the intensity of urban carbon emissions. Finally, the development of the digital economy accelerates the spillover of green knowledge among different cities through network externalities [44], and promotes the reduction of urban carbon emission intensity. The development of the digital economy provides more information and information exchange channels for cities, which promotes the spillover of green knowledge among cities and further enhance the green innovation level of different cities. Green innovation can change the development mode of cities relying on high input and high pollution into a development mode relying more on innovation input and output, thereby reducing the intensity of urban carbon emissions.

In addition, the higher the level of economic development, the higher the economic complexity. The increase of economic complexity further increases the difficulty of reasonable allocation of resources, which may lead to more serious problems of ineffective resource allocation. The development of the digital economy provides a more timely and broader path for resource allocation within and between cities through big data, cloud platforms and other information collection and exchange technologies. Therefore, the development of the digital economy provides a basis for the improvement of urban resource allocation efficiency, and the improvement of urban resource efficiency further provides space for urban green innovation.

Hypothesis 3: *The digital economy is conducive to improving the level of urban green innovation, thereby promoting the reduction of urban carbon emission intensity.*

3. Variables, Data and Research Design

3.1. Variable Selection and Descriptive Statistics

Explained variable: urban carbon emission intensity (*CO*). In this paper, carbon dioxide emissions divided by real GDP are used as a measure of urban carbon emission intensity. For economies in transition, it is necessary to maintain economic growth and protect the ecological environment. Therefore, taking carbon emission intensity as the emission reduction indicator is based on the full consideration of China's economic and social development.

Core explanatory variable: urban digital economy level (*Dig*). Based on the method of Zhao et al. [45], this paper constructs an indicator system to measure the development level of the digital economy in cities from the two levels of internet development and digital financial inclusion, and uses the entropy method to calculate the development index of urban digital economy. In addition, this paper also uses the urban digital economy index released by Tencent Research Institute to test the robustness of replacing core explanatory variables.

In order to more accurately evaluate the impact of urban digital economy on carbon emission intensity, the following control variables are added: population size (*Peo*), industrial structure (*Str*), human capital (*Stu*), financial development (*Mon*), and foreign direct investment (*FDI*). In addition, this paper uses the year fixed effect to control the variable that affects the urban carbon emission intensity over year, and also controls the urban fixed effect to alleviate the endogenous problem caused by the error of missing variables. The endogenous test part also further adds the provincial fixed effect. We further add the lag phase I of urban carbon emission intensity as the control variable into the regression equation to reduce the endogenous problem of the model caused by missing variables. The standard errors in the regression process of this paper are all clustered to the enterprise level (except for the panel Tobit model).

3.2. Sample Selection and Data Sources

The sample of urban panel data used in this paper is from the Statistical Yearbook of Chinese Cities from 2012 to 2020, which excludes prefecture-level cities in the Tibet Autonomous Region, Hong Kong Special Administrative Region, Macao Special Administrative Region and Taiwan Province, and deletes the prefecture-level cities established and cancelled in 2013 and later, totaling 278 prefecture-level cities and above. In the robustness

test, the digital economy index calculated by replacing the entropy method with the urban digital economy index of Tencent Research Institute is used for regression. The time range of the urban digital economy index of Tencent Research Institute is from 2015 to 2019. Descriptive statistical results of variables are shown in Table 1.

Variable Type	Variable Definition	Variable Definition	Mean Value	Standard Deviation
Interpreted variable	Carbon emission intensity (CO)	Urban carbon emissions/urban GDP	0.4605	1.8569
Explanatory variable	Digital Economy (Dig)	Development level of urban digital economy	0.0949	0.0553
	Population size (Peo)	Total urban population at the end of the year	5.8874	0.6975
Control	Industrial structure (Str)	Output value of secondary industry/urban GDP	3.8190	0.2590
variable	Human capital (Stu)	Number of students in primary and secondary schools	3.3635	11.2038
	Financial Development (Mon)	Loan balance of financial institutions at the end of the year	3199.5065	6647.4130
	Foreign Direct Investment (FDI)	Actual utilization of foreign direct investment	10.0947	1.8665

Table 1. Variable definition and descriptive statistical results.

Source: the data is collected by the author.

3.3. Basic Regression Model Setting

This paper mainly studies the impact of digital economy development on urban carbon emission intensity. In addition to adding control variables, it also controls the fixed effects of time (year) and region (city). The regression model is as follows:

$$CO_{i,t} = \alpha + \beta Dig_{i,t} + \varphi X_{i,t} + \gamma_t + \delta_i + \varepsilon_{i,t}$$
(1)

In Equation (1), $CO_{i,t}$ are the explained variables in this paper, representing the carbon emission intensity of the *i* city in year *t*. $Dig_{i,t}$ is the key explanatory variable of this paper, representing the development level of the digital economy of city *i* in year *t*. β is the coefficient that we are most concerned about, indicating the impact of the development of the urban digital economy on the carbon emission intensity. If the regression result is significantly positive, it indicates that the development of the urban digital economy has increased the carbon emission intensity. If the regression result is significantly negative, it indicates that the development of the urban digital economy has reduced the carbon emission intensity. $X_{i,t}$ is the control variable, specifically: population size (*Peo*), industrial structure (*Str*), human capital (*Stu*), financial development (*Mon*), foreign direct investment (*FDI*). α is a constant term. $\varepsilon_{i,t}$ is the random error term. γ_t and δ_i are year and city fixed effects, respectively.

4. Basic Regression Results

This paper first empirically studies the impact of the development of the urban digital economy on carbon emission intensity, and then conducts a robustness test from four aspects to verify the reliability of the basic regression results, followed by an endogenous test.

4.1. Analysis of Basic Results

The research on the impact of digital economy development on urban carbon emission intensity takes Equation (1) as the empirical research model, and the basic regression results are shown in Table 2. The explained variable is the carbon emission intensity of cities, and the core explanatory variable is the digital economy development level of cities at prefecture level and above. Model 1 in column (1) of Table 2 does not add control variables and the fixed effects of year and city. The regression coefficient of the digital economy is -0.4138, passing the test at the 1% significance level. The regression results show that the

development of the digital economy has significantly promoted the reduction of urban carbon emission intensity. The second model in column (2) of Table 2 is based on model 1, adding year and urban fixed effects. The regression coefficient of the digital economy is -1.1451, passing the test at the 1% significance level. We are most concerned about the regression results of model 3 in column (3) of Table 2. Model 3 adds control variables on the basis of model 2, and controls the year and urban fixed effects. The regression coefficient of the digital economy is -0.3780, passing the test at the 5% significance level. The regression results show that the development of the digital economy has significantly promoted the reduction of urban carbon emission intensity.

Explained Variable: Urban Carbon Emission Intensity	(1) Model I	(2) Model II	(3) Model III
Disital company	-0.4138 ***	-1.1451 ***	-0.3780 **
Digital economy	(0.1591)	(0.3620)	(0.1799)
Constant	0.4212 ***	0.0617	2.5501 **
Constant	(0.0152)	(0.5187)	(1.0373)
Control variable	No	No	Yes
Year fixed effect	No	Yes	Yes
Urban fixed effect	No	Yes	Yes
Obs.	2395	2395	2395
Within R ²	0.3686	0.0587	0.3693
Number of cities	278	278	278

Table 2. Basic regression results.

Notes: ***, ** are significant at the level of 1% and 5%, respectively. The values in parentheses are the standard errors of variable estimation coefficients clustered by cities.

The regression results from column (1) to (3) show that, under the condition that other conditions remain unchanged, the development of the digital economy has significantly promoted the decline of urban carbon emission intensity, and the development of the digital economy has an important role in promoting the green transformation of urban economy. Hypothesis 1 of this paper has been verified. The research conclusions of this paper are similar to those of Zhu et al. [46], which promote the sustainable development of the environment [47].

4.2. Robustness Test

The test results in Table 2 show that the development of the digital economy can effectively promote the reduction of urban carbon emission intensity. However, the robustness of the basic regression results has not been tested. This paper further verifies the robustness of the basic regression results by transforming core explanatory variables, deleting municipal samples, using panel Tobit model, and censoring and shrinking tail tests. In the process of robustness test based on Equation (1), the control variables, year and urban fixed effects are added, and the clustering robustness standard error at the urban level is adopted.

4.2.1. Change the Measurement Method of Core Explanatory Variables

In the basic regression process of this paper, the core explanatory variable adopts the entropy method to measure the development level of the urban digital economy. In order to verify the robustness of the basic regression results, this paper uses the urban digital economy index released by Tencent Research Institute as the core explanatory variable for regression again. The regression results are shown in column (1) of Table 3. The regression coefficient of the digital economy is -0.0095. The test at the 10% significance level shows that the development of the digital economy has significantly reduced the intensity of urban carbon emissions. The basic regression results in Table 2 do not change due to the change in the measurement of core explanatory variables, and the basic conclusions are robust.

	(1)	(2)	(3)	(4)	(5)	(6)
Explained Variable: Urban Carbon Emission Intensity	Tencent Digital Economic Index	Number of Internet Users	Eliminate Municipality Directly under the Central Government	Tobit Model	Shrinkage Test	Censoring Test
Digital Economy a	-0.0095 * (0.0050)					
Digital Economy b		-0.0409 * (0.0240)				
Digital economy			-0.3660 *	-0.0961 **	-0.2915 ***	-0.2915 ***
			(0.1999)	(0.0490)	(0.1087)	(0.1002)
Control variable	Yes	Yes	Yes	Yes	Yes	Yes
Year fixed effect	Yes	Yes	Yes	Yes	Yes	Yes
Urban fixed effect	Yes	Yes	Yes	Yes	Yes	Yes
Constant	0.2083 (2.9518)	3.1014 *** (1.1255)	2.5915 ** (1.0403)	4.8172 *** (1.2940)	2.5759 *** (0.6345)	2.8754 *** (0.5841)
Obs.	1037	2390	2359	2395	2395	2283
Within R ²	0.394	0.3697	0.3694	_	0.8267	0.8359

Table 3. Results of robustness test.

Notes: ***, **, * are significant at the level of 1%, 5% and 10%, respectively. The values in parentheses are the standard errors of variable estimation coefficients clustered by cities. Digital Economy a is the urban digital economy index released by Tencent Research Institute. Digital Economy b is the number of internet users.

In addition, this paper also uses the number of internet users as a variable to measure the development level of the urban digital economy, and puts it into the regression equation to test again. The regression results are shown in in the regression results in column (2) of Table 3, the regression coefficient of the digital economy is -0.0409, passing the test at the 10% significance level. It still shows that the development of the digital economy can reduce the carbon emission intensity of cities. The basic regression results are robust.

4.2.2. Delete Samples from Municipalities Directly under the Central Government

Considering that the administrative levels of municipalities and provinces (autonomous regions) are the same, this paper further deletes the samples of municipalities for another regression. The regression results of sample deletion of municipalities are shown in Table 3. The regression in column (3) of Table 3 shows that the regression coefficient of the digital economy is -0.3660, which passes the test at the significance level of 10%. The digital economy promotes the reduction of urban carbon emission reduction intensity, and there is no different result due to the change in sample selection. The basic regression result is reliable.

4.2.3. Panel Tobit Model Regression

Because the urban carbon emission intensity data in this paper has the left deletion problem with 0 as the critical point, this paper further uses the panel Tobit model for regression. The regression results are shown in column (4) of Table 3. The regression coefficient of the digital economy is -0.0961, which passes the test at the significance level of 5%. The regression results still show that the development of the digital economy has promoted the decline in urban carbon emission intensity. The conclusions of this paper could not change due to the transformation of regression methods, so the basic regression results are robust.

4.2.4. Shrinking and Ending Inspection

Due to measurement error and other reasons, the sample data could be biased to a certain extent, thus affecting the objectivity and accuracy of the regression results. Therefore, for the possible extreme values and other special cases, the sample can generally be shrunk and truncated. In this paper, we choose 1.5 and 97.5 percentiles for sample shrinkage, and lower than the 1st percentile and higher than the 99th percentile for sample truncation. The

regression results are shown in Table 3. The column (5) of Table 3 is the regression result of the shrinkage test. The regression coefficient of the digital economy is -0.2915, which passes the test at the 1% significance level. The column (6) of Table 3 is the regression result of the truncation test. The regression coefficient of the digital economy is -0.2915, which also passes the test at the 1% significance level. The results of tail shrinking and censoring tests show that the basic regression results of this paper are reliable, and the development of the digital economy has significantly reduced the intensity of urban carbon emissions.

4.3. Endogenetic Test

In this paper, the endogeneity test is conducted from two aspects: sample selectivity bias and missing variables.

4.3.1. Sample Selection Deviation Test

Because different cities have different stages of economic development and different pillar industries, not all cities' carbon emission intensity could decline, so this paper further adopts Heckman two-step method to test. Specifically, the Heckman two-step method uses the Probit model to calculate the inverse Mills ratio (IMRatio) in the first step, and brings the calculated inverse Mills ratio into the basic regression equation for re regression in the second step. In the first step of calculation, this paper calculates the growth rate of carbon emission intensity of each city. If the growth rate is greater than zero, it is taken as 1; if it is less than zero, it is taken as 0. Three regression methods are used in the Heckman two-step method in this paper.

The first way is to select the control variables in the basic regression as the influence variables of the first stage Probit model in the sample selection model. The test results of Heckman two-step method in this way are shown in column (1) of Table 4. The regression coefficient of the digital economy is -0.4613, which passes the test at the 5% significance level. The estimated coefficient of inverse Mills ratio is 6.6193, which fails the significance test. This indicates that the measurement process in this paper does not have the problem of sample selectivity error.

	(1)	(2)	(3)	(4)	(5)	(6)
	-0.4613 **	-0.3422 *	-0.3653 **	-0.3780 **	-0.4144 **	-0.4144 **
Digital economy	(0.2203)	(0.1768)	(0.1737)	(0.1810)	(0.2018)	(0.2032)
	6.6193	-1.8977	-0.8665			
IMRatio	(5.8118)	(1.3038)	(0.6366)			
	8.8624	4.4803 ***	1.6815	2.5501 **	2.8825 **	2.8825 **
Constant	(5.7697)	(1.3132)	(1.4472)	(1.0437)	(1.2135)	(1.2219)
Control variable	Yes	Yes	Yes	Yes	Yes	Yes
Year fixed effect	Yes	Yes	Yes	Yes	Yes	Yes
Urban fixed effect	Yes	Yes	Yes	Yes	Yes	Yes
Provincial fixed effect	Yes	Yes	Yes	Yes	No	Yes
Obs.	2392	2122	2122	2395	2124	2124
Within R ²	0.0020	0.0026	0.0025	0.3693	0.3653	0.3653

Table 4. Endogenous test results.

Notes: ***, **, * are significant at the level of 1%, 5% and 10%, respectively. The values in parentheses are the standard errors of variable estimation coefficients clustered by cities.

In the second way, only the lag period of carbon emission intensity is selected as the influence variable of the Probit model in the first stage of the sample selection model. The test results of Heckman two-step method in this way are shown in column (2) of Table 4. The regression coefficient of the digital economy is -0.3422, which passes the test at the 10% significance level. The estimated coefficient of inverse Mills ratio is -1.8977, which fails the significance test. This indicates that there is no sample selectivity error in the measurement process in this paper.

The third method is the integration of the first and second methods. It not only adds all control variables, but also adds the lag phase I of carbon emission intensity as the impact variable of the first phase Probit model in the sample selection model. The test results of Heckman two-step method in this way are shown in column (3) of Table 4. The regression coefficient of the digital economy is -0.3653, which passes the test at the 5% significance level. The estimation coefficient of inverse Mills ratio is -0.8665, which fails the significance test. This indicates that there is no sample selectivity error in the measurement process in this paper.

In conclusion, the Heckman two-step method shows that the regression samples in this paper cannot have the problem of selective bias, and the development of the digital economy can promote the reduction of urban carbon emission intensity.

4.3.2. Missing Variables

In this paper, the control variables at the city level and the fixed effects of city and year are added to the basic regression, which reduces the endogenous problems caused by missing variables to a certain extent. However, there may still be missing variables due to time and regional changes in different regions. Therefore, this paper further adds provincial fixed effects and urban carbon emission intensity lags into the regression equation to further reduce the endogenous problems caused by missing variables. Specifically, column (4) of Table 4 adds provincial fixed effects, and the regression coefficient of the digital economy is -0.3780. It passes the test at the 5% significance level, indicating that after adding provincial fixed effects, the digital economy still significantly reduces the intensity of urban carbon emissions. In column (5) of Table 4, the lag phase I of urban carbon emission intensity is added as the control variable for regression again, without controlling the fixed effect of provinces. The regression results show that the regression coefficient of the digital economy is -0.4144, passing the test at the significance level of 5%, which still shows that the digital economy has promoted the reduction of urban carbon emission intensity. In column (6) of Table 4, the lag phase of urban carbon emission intensity is added, and the fixed effect of provinces is controlled. In the regression results, the regression coefficient of the digital economy is -0.4144. Through the test at the significance level of 5%, the improvement of the development level of the digital economy can effectively reduce the urban carbon emission intensity. Therefore, the basic research conclusions of this paper are reliable.

5. Mechanism Inspection

Through basic regression, robustness test and endogenous test, this paper has confirmed that the development of the digital economy can promote the reduction of urban carbon emission intensity, but how the digital economy plays this role is still worth further exploring. The regression equation of mechanism research in this paper is defined in Equation (2):

$$Jizhi_{i,t} = \alpha + \mu Dig_{i,t} + \varphi X_{i,t} + \gamma_t + \delta_i + \varepsilon_{i,t}$$
⁽²⁾

In Equation (2), $Jizhi_{i,t}$ represents the mechanism variables in the regression analysis of this paper, which are the human capital investment and green innovation level of city *i* in year *t*, respectively. μ is the coefficient that we are most concerned about, and the other variables are the same as Equation (1). Human capital investment is expressed in urban education expenditure. The level of urban green innovation is expressed by urban per capita green patent data. Urban green patent data include green inventions and green utility models. The education expenditure is from the China Urban Statistical Yearbook from 2012 to 2020, and the urban green patent data is from CNRDS. In the regression process, all control variables are added and the city and year effects are fixed, and the standard error is clustered to the city level.

5.1. Improvement of Human Capital Investment

Firstly, this paper discusses the human capital investment mechanism that the digital economy promotes the reduction of urban carbon emission intensity. According to Equation (2), this paper uses human capital investment to regress digital economic variables in empirical analysis. The test results of human capital investment enhancement mechanism are shown in column (1) of Table 5. The regression coefficient of the digital economy is 0.4715, which passes the test at the significance level of 1%. It shows that the development of the digital economy has significantly promoted the improvement of urban human capital investment, and then led to the reduction of urban carbon emission intensity. Hypothesis 2 is validated.

Table 5. Mechanism inspection results.

	(1) Human Capital Investment	(2) Green Innovation Level
Digital aconomy	0.4715 ***	1.9280 ***
Digital economy	(0.1535)	(0.5331)
Constant	5.6600 ***	-2.3434
Constant	(1.0044)	(1.6156)
Control variable	Yes	Yes
Year fixed effect	Yes	Yes
Urban fixed effect	Yes	Yes
Obs.	2395	2389
Within R ²	0.1960	0.2177

Notes: *** is significant at the level of 1%. The values in parentheses are the standard errors of variable estimation coefficients clustered by cities.

5.2. Improve the Level of Green Innovation

The test results of the mechanism for improving the level of green innovation are shown in column (2) of Table 5. The regression coefficient of the digital economy is 1.9280, passing the test at the significance level of 1%. It shows that the development of the digital economy has significantly improved the level of urban green innovation, and the improvement of the level of green innovation has promoted the decline of urban carbon emission intensity. The higher the level of green innovation, the lower the intensity of urban carbon emissions. The level of green innovation is an important factor affecting the intensity of carbon emissions. The development of the digital economy can not only promote the improvement of urban green innovation level [48], but also promote the enhancement of enterprises' green technology capability and the increase of enterprises' green patents [49]. Hypothesis 3 of this paper is verified.

6. Further Analysis

This paper further analyzes the different classifications of urban location (eastern, central and western regions), urban administrative level, urban population size, resourcebased cities and non-resource-based cities, with a view to studying the heterogeneity effect of digital economy development on urban carbon emission intensity and whether the mechanism of human capital investment and green innovation level is still in place. In the further analysis, all control variables are added based on Equations (1) and (2) to control the year and urban fixed effect, and the standard error of regression coefficient is clustered to the urban level.

6.1. East, Middle and West China

The regression results of the influence of urban digital economy development level on carbon emission intensity in the eastern, central and western regions are shown in Table 6. Column (1) lists the regression results of urban samples in the eastern region. The results show that the regression coefficient of the urban digital economy development level in the eastern region is -0.5800. The test at the 5% significance level shows that the improvement of the urban digital economy development level in the eastern region can effectively promote the reduction of the urban carbon emission intensity in the eastern region. Column (2) shows the regression results of the urban sample in the central region. The results show that the regression coefficient of the development level of the urban digital economy in the central region is -0.5182. The test at the 10% significance level shows that the development of the urban digital economy in the central region has significantly reduced the intensity of carbon emissions. Column (3) presents the regression results of the urban sample in the western region. The regression coefficient of the digital economy is 0.2273, which does not pass the significance test. To sum up, the regression results in Table 6 suggest that the development of the urban digital economy in eastern and central can promote the reduction of carbon emission intensity, but there is no significant effect in western China.

	(1) Eastern Region	(2) Central Region	(3) Western Region
Digital aconomy	-0.5800 **	-0.5182 *	0.2273
Digital economy	(0.2912)	(0.2864)	(0.5303)
	9.6363 **	1.8999 ***	-0.6146
Constant	(4.0435)	(0.5752)	(1.9841)
Control variable	Yes	Yes	Yes
Year fixed effect	Yes	Yes	Yes
Urban fixed effect	Yes	Yes	Yes
Obs.	881	868	646
Within R ²	0.2349	0.8781	0.9147

Table 6. Regression results of urban samples in eastern, central and western China.

Notes: ***, **, * are significant at the level of 1%, 5% and 10%, respectively. The values in parentheses are the standard errors of variable estimation coefficients clustered by cities. The explained variables are the urban carbon emission intensity.

The possible reasons are that on the one hand, compared with the western region, the eastern and central regions have a higher level of urban economic development, and the eastern and central regions have a higher degree of marketization, which is more conducive to the formation and development of the digital economy. The industrial agglomeration and economies of scale produced by the formation and development of the digital economy are more significant, and the development of the digital economy is more conducive to the improvement of the efficiency of urban resource allocation. On the other hand, the cities in the eastern and central regions have relatively early economic transformation and upgrading, while the cities in the western regions have relatively lagged behind in transformation and upgrading. The regions with early economic transformation have started to lay out the infrastructure that is conducive to the development of the digital infrastructure has formed economies of scale and played an economic role, while the regions with late transformation have limited the role of the digital economy.

Table 7 shows the test results of the influence mechanism of urban digital economy development level on carbon emission intensity in the eastern and central regions. Column (2) reports the regression results of the human capital investment mechanism of urban samples in the eastern region. The results show that the regression coefficient of the urban digital economy development level in the eastern region is 0.3796, passing the test at the significance level of 5%, indicating that the development of the digital economy has promoted the improvement of human capital investment in the eastern region, thereby reducing the carbon emission intensity of cities in the eastern region. Column (2) lists the regression results of the green innovation level mechanism of urban samples in the eastern region. The results show that the regression coefficient of the urban digital economy development level in the eastern region. The results show that the regression coefficient of the urban digital economy development level in the eastern region. The results show that the regression coefficient of the urban digital economy development level in the eastern region is 1.2854, passing the test at the 5% significance level, indicating that the development of the digital economy has promoted the improvement of the green innovation level in the eastern region, thereby reducing the carbon emission intensity of

cities in the eastern region. Column (3) presents the regression results of the human capital investment mechanism of the city samples in the central region. The results show that the regression coefficient of the urban digital economy development level in the central region is 0.7794, passing the test at the 5% significance level, indicating that the urban digital economy development in the central region significantly reduces the carbon emission intensity through the increase of human capital investment. Column (4) shows the regression results of the green innovation level mechanism of urban samples in the central region. The results show that the regression coefficient of the urban digital economy development level in the central region is 1.3732, passing the test at the 10% significance level, indicating that the development of the urban digital economy in the central region has significantly reduced the carbon emission intensity by improving the green innovation level.

	(1)	(2)	(3)	(4)
	Eas	tern Region	Cer	ntral Region
	Human Capital Investment	Green Innovation Level	Human Capital Investment	Green Innovation Level
Digital economy	0.3796 **	1.2854 **	0.7794 **	1.3732 *
	(0.1461)	(0.4935)	(0.3559)	(0.7392)
_	3.5412 *	-12.9212 ***	6.6639 ***	-0.0839
Constant	(1.8263)	(3.0088)	(1.1598)	(0.4155)
Control variable	Yes	Yes	Yes	Yes
Year fixed effect	Yes	Yes	Yes	Yes
Urban fixed effect	Yes	Yes	Yes	Yes
Obs.	881	880	868	866
Within R ²	0.2536	0.3142	0.2402	0.1747

Table 7. Mechanism test results of east and central China.

Notes: ***, **, * are significant at the level of 1%, 5% and 10%, respectively. The values in parentheses are the standard errors of variable estimation coefficients clustered by cities.

6.2. Urban Administrative Grade

Due to the different administrative levels among Chinese cities, the difference of resource gathering capacity is often caused. Cities at sub provincial level and above could receive more resource allocation, more policy preference and capital investment. Provincial capital cities and ordinary prefecture level cities gradually take the second place. In particular, ordinary prefecture level cities are often attracted by cities above the provincial capital level, and most of the resource-oriented cities or ordinary prefecture level cities are even more lacking in technology, capital and human capital.

Table 8 shows for the regression results of the impact of the urban digital economy development level of different administrative levels on carbon emission intensity. Column (1) lists the regression results of the sample of cities at and above the sub-provincial level. The results show that the regression coefficient of the development level of the digital economy in cities at and above the sub-provincial level is -0.2208. The test at the significance level of 5% shows that the development of the digital economy in cities at and above the sub-provincial level has reduced the intensity of urban carbon emissions. Column (2) shows the regression results of the provincial capital city sample. The results show that the regression coefficient of the provincial capital city's digital economic development level is 0.2824, which fails the significance test. Column (3) displays the regression results of other cities. The regression coefficient of the digital economy is -0.4815, which fails the significance test. In short, the regression results in Table 8 show that China's urban administrative level could affect the effect of digital economy development on urban carbon emission intensity. The development of the digital economy in cities at sub provincial level and above can significantly promote the reduction of urban carbon emission intensity, but the impact of digital economy development on carbon emission intensity in provincial capital cities and other cities does not show a significant effect.

	(1)	(2)	(3)	(4)	(5)
	Cities above Sub Provincial Level	Provincial Capital	Other Cities	Cities above Sub I Human Capital Investment	Provincial Level Green Innovation Level
Digital aconomy	-0.2208 **	0.2824	-0.4815	0.6008 **	0.9113 *
Digital economy	(0.1018)	(0.4704)	(0.3373)	(0.2601)	(0.4721)
Constant	3.7417	4.2284	2.4147 **	-5.5308	-11.9261
Constant	(2.3203)	(6.0485)	(1.1468)	(3.6063)	(7.0629)
Control variable	Yes	Yes	Yes	Yes	Yes
Year fixed effect	Yes	Yes	Yes	Yes	Yes
Urban fixed effect	Yes	Yes	Yes	Yes	Yes
Obs.	159	132	2104	159	159
Within R ²	0.8543	0.4677	0.3664	0.5266	0.0935

Table 8. Regression results of samples of cities with different administrative levels.

Notes: **, * are significant at the level of 5% and 10%, respectively. The values in parentheses are the standard errors of variable estimation coefficients clustered by cities.

Therefore, the impact of digital economy development on urban carbon emission intensity varies with the different administrative levels of cities. The possible reasons are that, on the one hand, cities at sub provincial level and above have more preferential policies and stronger strength, so they are the first to receive national policy support in the development of the digital economy, and for their own province or city, the government could also use stronger strength to support their first development of the digital economy. Therefore, the development level and stage of the digital economy are higher. On the other hand, the marketization level of cities at or above the sub provincial level is among the best in domestic cities. Both the government business environment and the market resource allocation capacity are stronger than other cities. Therefore, the development of the digital economy is more conducive to improving urban environmental performance. Therefore, the promotion effect of the digital economy on the reduction of carbon emission intensity in sub provincial and above cities is more obvious, and these cities have also obtained the positive effect of the digital economy on environmental governance earlier.

The sample mechanism test results of cities above sub-provincial level are also shown in Table 8. Columns (4) and (5) are the mechanism test results of the human capital investment and green innovation level of the cities above the sub provincial level. The results show that the regression coefficients of the digital economy of cities above the sub-provincial level are 0.6008 and 0.9113, respectively, which have passed the test at the significance level of 5% and 10%, respectively, indicating that in cities above the sub-provincial level, the development of the digital economy has promoted the increase of human capital investment and the improvement of the level of green innovation, thus reducing the intensity of carbon emissions.

6.3. Urban Population Size

In addition to the differences in location and administrative level, there are also differences in city size. We divide cities with a population of more than 5 million into big cities, medium-sized cities between 1 million and 5 million, and small cities less than 1 million. We further analyze the impact of the digital economy on carbon emission intensity in different cities. The regression results are shown in Table 9. Column (1) lists the regression results of the sample of large cities. The results show that the regression coefficient of the development level of the digital economy in the sample of large cities is -0.5603. The test at the significance level of 5% shows that the development of the digital economy can promote the reduction of carbon emission intensity in large cities. Column (2) shows the regression results of the sample of medium-sized cities. The regression coefficient of the digital economy of medium-sized cities is -0.0747, which failed the significance test. Column (3) presents the regression results of small city samples. The regression coefficient of the digital economy is 2.0203, which fails the significance test. In a word, the

regression results in Table 9 show that the development of the digital economy in large cities has promoted the reduction of carbon emission intensity, but in the sample of small and medium-sized cities, the development of the digital economy has not promoted the reduction of urban carbon emission intensity.

- 1 • 1 • 7 • 1 1	(1)	(2)	(3)	(4)	(5)	
Explained Variable:		26 11		Big City		
Urban Carbon Emission Intensity	Big City	Medium- Sized City	Small City	Human Capital Investment	Green Innovation Level	
Digital economy	-0.5603 **	-0.0747	2.0203	0.5435 ***	2.2792 ***	
	(0.2767)	(0.1544)	(2.0893)	(0.1961)	(0.7123)	
_	-0.2217	4.6597 ***	2.354	6.7183 ***	-5.1531 ***	
Constant	(3.2434)	(1.5960)	(5.2189)	(1.5780)	(1.8703)	
Control variable	Yes	Yes	Yes	Yes	Yes	
Year fixed effect	Yes	Yes	Yes	Yes	Yes	
Urban fixed effect	Yes	Yes	Yes	Yes	Yes	
Obs.	907	1418	70	905	903	
Within R ²	0.1952	0.6198	0.9376	0.1096	0.2821	

Table 9. Regression results of large, medium and small cities.

Notes: ***, ** are significant at the level of 1% and 5%, respectively. The values in parentheses are the standard errors of variable estimation coefficients clustered by cities.

Table 9 shows the mechanism test results of the sample of large cities. Columns (4) and (5) are the mechanism test results of human capital investment and green innovation level of large cities, respectively. The results show that the regression coefficients of the digital economy in big cities are 0.5435 and 2.2792, respectively, which pass the test at the significance level of 1%. It shows that the development of the digital economy in big cities has promoted the increase of their human capital investment and the improvement of their green innovation level, thereby reducing their carbon emission intensity.

6.4. Resource-Based Cities and Non-Resource-Based Cities

The transformation and development of resource-based cities has played a vital role in the healthy development of China's economy and society. Most resource-based cities take the mining and processing of mineral resources with high pollution and high energy consumption as their pillar industries. As early as 2007, the State Council issued Several Opinions of the State Council on Promoting the Sustainable Development of Resource based Cities, which defined that resource-based cities should strengthen environmental politics and ecological protection. However, no matter the economic and social transformation [50] and development of resource-based cities, or the ecological environment protection, the effect is not obvious. Industrial upgrading is an important means to balance economic development and environmental protection in resource-based cities. The development of the digital economy may provide an effective way for resource-based cities to achieve the unification of "green water and green mountains" and "golden mountains and silver mountains", and open the key links of industrial upgrading of resource-based cities.

Table 10 shows the regression results of the impact of the development level of the digital economy of resource-based cities and non-resource-based cities on carbon emission intensity. Column (1) of Table 10 shows the regression results of the sample of resource-based cities. The regression coefficient of the development level of the digital economy is -0.3225, which fails the significance test. Column (2) of Table 10 shows the regression results of the sample of non-resource cities. The regression coefficient of the digital economy is -0.3632, which passes the test at the significance level of 5%. Therefore, the digital economy promotes the decline of urban carbon emission intensity and plays a more significant role in non-resource cities. The test results of carbon emission reduction mechanism for non-resource-based cities are shown in columns (3)–(4) of Table 10. Columns (3) and (4) of Table 10 are the mechanism tests of human capital investment and green innovation level,

respectively. The regression coefficients of the digital economy in the regression results are 0.3329 and 1.6160, respectively, passing the test at 5% and 1% significance levels. It shows that in non-resource cities, the development of the digital economy promotes the reduction of urban carbon emission intensity by increasing human capital investment and improving the level of green innovation.

	(1)	(2)	(3)	(4)	
		Non-Resource-	Non-Resource-Based Cities		
	Resource-Based City	Based Cities	Human	Green	
			Capital Investment	Innovation Level	
Digital aconomy	-0.3225	-0.3632 **	0.3329 **	1.6160 ***	
Digital economy	(0.8891)	(0.1731)	(0.1320)	(0.5436)	
	2.1620 **	3.0900 *	6.2054 ***	-4.8307 **	
Constant	(1.0745)	(1.8421)	(1.0627)	(2.3030)	
Control variable	Yes	Yes	Yes	Yes	
Year fixed effect	Yes	Yes	Yes	Yes	
Urban fixed effect	Yes	Yes	Yes	Yes	
Obs.	943	1452	1452	1449	
Within R ²	0.3362	0.4466	0.1883	0.2635	

Table 10. Regression results of resource-based cities and non-resource-based cities.

Notes: ***, ** are significant at the level of 1% and 5%, respectively. The values in parentheses are the standard errors of variable estimation coefficients clustered by cities.

For resource-based cities and non-resource-based cities, the development of the digital economy in non-resource-based cities can more effectively reduce carbon emission intensity. China's resource-based cities have different classifications. The National Sustainable Development Plan for Resource-based Cities (2013–2020) divides resource-based cities into four types: growth, maturity, recession and regeneration. Does the digital economy not promote the reduction of urban carbon emission intensity among the four types of resource-based cities? It is worth further study.

Regression results of the impact of the digital economy on urban carbon emission intensity in four different types of resource-based cities are shown in Table 11. Column (1) shows the regression results of the sample of growth resource-based cities. The regression coefficient of the digital economy development level is 0.8214, which fails the significance test. Column (2) reports the regression results of the sample of mature resource-based cities. The regression coefficient of the digital economy development level is 1.0426, which fails the significance test. Column (3) displays the regression results of the sample of declining resource-based cities. The regression coefficient of the digital economy development level is 0.6427, which fails the significance test. Column (4) shows the regression results of the sample of renewable resource-based cities. The regression coefficient of the digital economy is -0.6057, which passes the test at the level of 10% significance. To sum up, the digital economy promotes the reduction of urban carbon emission intensity, and plays a more significant role in renewable cities among non-resource-based cities. Therefore, in Table 10, the effect of the digital economy on the reduction of carbon emission intensity of resource-based cities is not significant, which is mainly due to the fact that the effect is not significant in growing, mature and declining resource-based cities.

In China's resource-based cities, they are mainly cities that are gradually established and emerging relying on the mining of coal, oil, forests and non-ferrous metal minerals. According to the different resources that resource-based cities mainly rely on, this paper divides resource-based cities into coal resource-based cities, oil resource-based cities, iron ore resource-based cities, forest resource-based cities and other mineral resource-based cities. To further explore the impact of the digital economy on carbon emissions of different resource-based cities.

	(1) Growth	(2) Mature Type	(3) Declination Type	(4) Regenerative Type
Distribution	0.8214	1.0426	0.6427	-0.6057 *
Digital economy	(0.5608)	(1.1299)	(0.8919)	(0.3424)
	7.1617 *	-1.0175	4.2807	31.0696 **
Constant	(3.8404)	(3.4389)	(2.8872)	(14.2426)
Control variable	Yes	Yes	Yes	Yes
Year fixed effect	Yes	Yes	Yes	Yes
Urban fixed effect	Yes	Yes	Yes	Yes
Obs.	99	523	190	131
Within R ²	0.9219	0.3105	0.9240	0.7429

Table 11. Regression results of samples of resource-based cities and non-resource-based cities.

Notes: **, * are significant at the level of 5% and 10%, respectively. The values in parentheses are the standard errors of variable estimation coefficients clustered by cities.

Table 12 shows the regression results of the impact of digital economy development level of different resource-oriented cities on carbon emission intensity. Column (1) shows the resource-based cities dominated by iron ore. The regression coefficient of the digital economy is -32.6671. Through the test at the significance level of 5%, it is shown that the development of the digital economy in resource-based cities dominated by iron ore has promoted the decline of urban carbon emission intensity. Column (2) is a resource-based city dominated by oil. The regression coefficient of the digital economy is -2.2687, passing the test at the 10% significance level, indicating that the development of the digital economy in resource-based cities dominated by oil also significantly promoted the reduction of urban carbon emission intensity. Columns (3)-(5) are resource-based cities dominated by coal, forest and other minerals. The regression coefficients of the digital economy are 2.1096, -0.1779 and 1.3966, respectively, which have not passed the significance test, indicating that the carbon intensity decline effect of urban digital economy development dominated by these three resources has not yet appeared. To sum up, the digital economy has significantly reduced the urban carbon emission intensity for resource-based cities dominated by iron ore and oil. In resource-based cities dominated by coal, forest and other minerals, the reduction of carbon emission intensity has not had a significant effect. Therefore, the digital economy of resource-based cities in Table 10 does not significantly promote the reduction of urban carbon emission intensity, which is more caused by the non-significant reduction effect of carbon emission intensity of coal, forest and other mineral resource-based cities.

Table 12. Regression results of samples of resource-based cities dominated by different industries.

	(1) Iron Ore	(2) Petroleum	(3) Coal	(4) Forest	(5) Other Minerals
Digital agan amy	-32.6671 **	-2.2687 ***	2.1096	-0.1779	1.3966
Digital economy	(13.2917)	(0.5239)	(1.9016)	(0.5792)	(0.9907)
<u> </u>	-9.5631	4.7608 **	1.1803	4.8001	1.6408
Constant	(21.9625)	(2.0404)	(3.2374)	(2.9640)	(2.9201)
Control variable	Yes	Yes	Yes	Yes	Yes
Year fixed effect	Yes	Yes	Yes	Yes	Yes
Urban fixed effect	Yes	Yes	Yes	Yes	Yes
Obs.	81	75	360	47	225
Within R ²	0.3090	0.9238	0.3567	0.9833	0.7894

Notes: ***, ** are significant at the level of 1% and 5%, respectively. The values in parentheses are the standard errors of variable estimation coefficients clustered by cities.

7. Conclusions

It is worth exploring whether the development of the digital economy can promote the protection and improvement of the ecological environment, and whether it can realize that "green water and green mountains are golden mountains and silver mountains". To some

extent, the reduction of carbon emission intensity can indicate that China has achieved the protection of ecological environment in the process of economic development. Therefore, this paper analyzes its impact on the decline of urban carbon emission intensity from the perspective of the digital economy. This paper first analyzes the theoretical basis and impact mechanism of the digital economy on urban carbon emission intensity, and then empirically analyzes the impact of the digital economy on urban carbon emission intensity and its mechanism using 278 city-level statistical data from 2011 to 2019. The research conclusions are as follows.

First, the development of the urban digital economy has significantly promoted the reduction of urban carbon emission intensity. In the process of the development of the digital economy, the production efficiency and government governance capacity have been improved, and the intensity of urban carbon emissions has decreased. After changing core explanatory variables, deleting municipality samples, using the panel Tobit model, and censoring and shrinking tail tests, the conclusion is still robust. Second, in terms of impact mechanism, the development of the urban digital economy, on the one hand, promotes the increase of urban human capital investment level, on the other hand, promotes the improvement of urban green innovation level, thereby reducing the intensity of urban carbon emissions. Third, the development of the urban digital economy has significantly promoted the reduction of carbon emission intensity in cities in the eastern and central regions, cities above the sub provincial level, large cities and non-resource-based cities. As far as resource-based cities are concerned, in renewable resource-based cities and resource-based cities and resource-based cities are concerned, in renewable resource-based cities and resource-based cities dominated by iron ore and oil mining, the effect of the digital economy on the reduction of urban carbon emission intensity is more obvious.

8. Policy Recommendations

Accordingly, the following policy recommendations are proposed. First, we should promote the development level of the urban digital economy. The development of the digital economy has promoted the reduction of urban carbon emission intensity. It shows that the development of the digital economy plays a positive role in ecological environment protection. Therefore, on the one hand, local governments should actively improve the infrastructure of the digital economy to lay a solid foundation for the development of the digital economy. On the other hand, entrepreneurs should give full play to their entrepreneurial spirit, promote the digital transformation of enterprises, and promote the development of the digital economy and carbon emission reduction at the enterprise level. Second, we could achieve the improvement of human capital investment and green innovation capability. The development of the digital economy has reduced the intensity of urban carbon emissions by promoting the level of urban human capital investment and green innovation. Therefore, local governments and enterprises should actively increase investment in human capital to make it more suitable for the development of the digital economy. Local governments and enterprises should also increase investment in R&D, especially in green innovation, to promote the improvement of production efficiency, ease the internal and external financing constraints of green innovation, and promote the improvement of green innovation level. Third, we should achieve balanced development of the digital economy. It can be seen from the empirical study that the carbon emission intensity reduction effect of the digital economy is more obvious in cities in the eastern and central regions, sub-provincial cities, large cities and non-resource-based cities. It shows that the development level of China's digital economy is not balanced. Therefore, on the one hand, local governments and enterprises should give full play to their own role to realize the development of the digital economy. On the other hand, they should give sufficient policy and financial support, and use the big data national strategy and the national computing network to achieve the balanced development of the digital economy. **Author Contributions:** Conceptualization, S.J., F.W., E.S. and M.D.; methodology, S.J., F.W. and M.D.; software, S.J.; validation, S.J., F.W. and X.W.; formal analysis, S.J., E.S. and M.D.; investigation, F.W.; resources, X.W.; data curation, S.J.; writing—original draft preparation, S.J., F.W. and X.W.; writing—review and editing, M.D.; visualization, S.J.; supervision, F.W.; funding acquisition, E.S. and M.D. All authors have read and agreed to the published version of the manuscript.

Funding: This research is funded by the Philosophy and Social Sciences Research of Higher Learning Institutions of Shanxi (PSSR) (2021W074), the National Natural Science Foundation of China (72003071), and the National Social Science Foundation of China (19BJY239).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are openly available in statistical yearbooks and official websites.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Balcerzak, A.P.; Pietrzak, M.B. Digital economy in visegrad countries. multiple-criteria decision analysis at regional level in the years 2012 and 2015. *J. Compet.* 2017, *9*, 5–18. [CrossRef]
- 2. Ali, M.A.; Hoque, M.R.; Alam, K. An empirical investigation of the relationship between e-government development and the digital economy: The case of Asian countries. *J. Knowl. Manag.* **2018**, *22*, 1176–1200. [CrossRef]
- 3. Bertani, F.; Ponta, L.; Raberto, M.; Teglio, A.; Cincotti, S. The complexity of the intangible digital economy: An agent-based model. *J. Bus. Res.* **2021**, *129*, 527–540. [CrossRef]
- 4. Mohamed, M.; Murray, A.; Mohamed, M. The role of information and communication technology (ict) in mobilization of sustainable development knowledge: A quantitative evaluation. *J. Knowl. Manag.* **2010**, *14*, 744–758. [CrossRef]
- 5. Noseleit, F. Entrepreneurship, structural change, and economic growth. J. Evol. Econ. 2013, 23, 735–766. [CrossRef]
- 6. Glaeser, E.L.; Kerr, S.P.; Kerr, W.R. Entrepreneurship and urban growth: An empirical assessment with historical mines. *Rev. Econ. Stat.* **2015**, *97*, 498–520. [CrossRef]
- 7. Jahanger, A. Impact of globalization on CO₂ emissions based on EKC hypothesis in developing world: The moderating role of human capital. *Environ. Sci. Pollut. Res.* **2022**, *29*, 20731–20751. [CrossRef]
- Lin, B.Q.; Huang, C.C. How will promoting the digital economy affect electricity intensity? *Energy Policy* 2023, 173, 113341. [CrossRef]
- 9. Cheng, Y.; Zhang, Y.; Wang, J.J.; Jiang, J.X. The impact of the urban digital economy on China's carbon intensity: Spatial spillover and mediating effect. *Resour. Conserv. Recycl.* 2023, 189, 106762. [CrossRef]
- 10. Zhang, G.; Zhang, N. Environmental regulation and worker benefits: Evidence from city-level air quality standards in China. *SSRN* **2020**, 3732620. [CrossRef]
- 11. Du, M.Z.; Feng, R.K.; Chen, Z.F. Blue sky defense in low-carbon pilot cities: A spatial spillover perspective of carbon emission efficiency. *Sci. Total Environ.* **2022**, *846*, 157509. [CrossRef]
- 12. Statistical Review of World Energy. Available online: https://www.bp.com/en/global/corporate/energy-economics/statistical-review-of-world-energy.html (accessed on 12 January 2023).
- 13. Ma, Q.; Khan, Z.; Tariq, M.; Hayriye, I.; Husam, R. Sustainable digital economy and trade adjusted carbon emissions: Evidence from China's provincial data. *Econ. Res.-Ekon. Istraživanja* **2022**, *35*, 5469–5485. [CrossRef]
- 14. Myovella, G.; Karacuka, M.; Haucap, J. Digitalization and economic growth: A comparative analysis of Sub-Saharan Africa and OECD economies. *Telecommun. Policy* **2020**, *44*, 101856. [CrossRef]
- 15. Zhao, Y.; Chen, X.H. The relationship between the withdrawal of the digital economy's innovators, government interventions, the marketization level and market size based on big data. *J. Enterp. Inf. Manag.* **2022**, *35*, 1202–1232. [CrossRef]
- Wallace, C. Can information and communications technology enhance social quality? *Int. J. Soc. Qual.* 2012, *2*, 98–117. [CrossRef]
 Chow, G.C. Capital formation and economic growth in China. *Q. J. Econ.* 1993, *108*, 809–842. [CrossRef]
- Brandt, L.; Van, B.J.; Zhang, Y. Creative accounting or creative destruction? Firm-level productivity growth in Chinese manufacturing. *J. Dev. Econ.* 2012, *97*, 339–351. [CrossRef]
- 19. Pan, W.R.; Xie, T.; Wang, Z.W.; Ma, L.S. Digital economy: An innovation driver for total factor productivity. *J. Bus. Res.* 2021, 139, 303–311. [CrossRef]
- 20. Vial, G. Understanding digital transformation: A review and a research agenda. J. Strateg. Inf. Syst. 2019, 28, 118–144. [CrossRef]
- 21. Goldfarb, A.; Tucker, C. Digital Economics. J. Econ. Lit. 2019, 57, 3–43. [CrossRef]
- 22. Yunis, M.; Tarhini, A.; Kassar, A. The role of ICT and innovation in enhancing organizational performance: The catalysing effect of corporate entrepreneurship. *J. Bus. Res.* **2018**, *88*, 344–356. [CrossRef]
- 23. Jiao, H.; Yang, J.F.; Wang, P.N.; Li, Q. Research on the mechanism of data driven dynamic capabilities of enterprises—Analysis of the digital transformation process based on data life-cycle management. *China Ind. Econ.* **2021**, *11*, 174–192.

- 24. He, P.S.; Lee, D.H. Evolution of the linkage structure of ICT industry and its role in the economic system: The case of Korea. *Inf. Technol. Dev.* **2019**, *25*, 424–454. [CrossRef]
- Ribeiro-Navarrete, S.; Botella-Carrubi, D.; Palacios-Marques, D.; Orero-Blat, M. The effect of digitalization on business performance: An applied study of KIBS. J. Bus. Res. 2021, 126, 319–326. [CrossRef]
- Autio, E.; Nambisan, S.; Thomas, L.D.W.; Wright, M. Digital affordances, spatial affordances, and the genesis of entrepreneurial ecosystems. *Strateg. Entrep. J.* 2018, 12, 72–95. [CrossRef]
- 27. Ladeira, M.J.M.; Ferreira, F.A.F.; Ferreira, J.J.M.; Fang, W.C.; Falcao, P.F.; Rosa, A.A. Exploring the determinants of digital entrepreneurship using fuzzy cognitive maps. *Int. Entrep. Manag. J.* **2019**, *15*, 1077–1101. [CrossRef]
- Hall, J.V.; Krueger, A.B. An analysis of the labor market for Uber's driver-partners in the United States. *ILR Rev.* 2018, 71, 705–732.
 [CrossRef]
- 29. Philip, L.J.; Townsend, L.; Roberts, E.; Beel, D. The Rural Digital Economy. Scott. Geogr. J. 2015, 131, 143–147. [CrossRef]
- 30. Gao, Y.; Zang, L.; Sun, J. Does computer penetration increase farmers' income? An empirical study from China. *Telecommun. Policy* **2018**, *42*, 345–360. [CrossRef]
- Syed, A.A.; Ahmed, F.; Muhammad, A.K.; Segovia, J.E.T. Assessing the role of digital finance on shadow economy and financial instability: An empirical analysis of selected South Asian countries. *Mathematics* 2021, 9, 3018. [CrossRef]
- Rochet, J.C.; Tirole, J. Tying in two-sided markets and the honor all cards rule. *Int. J. Ind. Organ.* 2008, *26*, 1333–1347. [CrossRef]
 Balsalobre-Lorente, D.; Driha, O.M.; Sinha, A. The dynamic effects of globalization process in analysing N-shaped tourism led
- growth hypothesis. *J. Hosp. Tour. Manag.* **2020**, *43*, 42–52. [CrossRef]
- Borsatto, J.M.L.S.; Amui, L.B.L. Green innovation: Unfolding the relation with environmental regulations and competitiveness. Resources. Conserv. Recycl. 2019, 149, 445–454. [CrossRef]
- 35. Wang, L.; Su, C.W.; Ali, S.; Chang, H.L. How China is fostering sustainable growth: The interplay of green investment and production-based emission. *Environ. Sci. Pollut. Res.* **2020**, *27*, 39607–39618. [CrossRef]
- Smulders, S.; Tsur, Y.; Zemel, A. Announcing climate policy: Can a green paradox arise without scarcity? J. Environ. Econ. Manag. 2012, 64, 364–376. [CrossRef]
- 37. Shapiro, J.S.; Walker, R. Why is pollution from u. s. manufacturing declining? The roles of environmental regulation, productivity, and trade. *Am. Econ. Rev.* **2018**, *108*, 3814–3854. [CrossRef]
- Tang, L.; Wu, J.; Yu, L.; Bao, Q. Carbon emissions trading scheme exploration in China: A multi-agent-based model. *Energy Policy* 2015, *81*, 152–169. [CrossRef]
- Khan, Y.; Hassan, T.; Tufail, M.; Marie, M.; Imran, M.; Zhang, X.Q. The nexus between CO₂ emissions, human capital, technology transfer, and renewable energy: Evidence from Belt and Road countries. *Environ. Sci. Pollut. Res.* 2022, 29, 59816–59834. [CrossRef]
- 40. Peters, M.A. Digital trade, digital economy and the digital economy partnership agreement (DEPA). *Educ. Philos. Theory* **2022**. [CrossRef]
- Li, J.; Liu, X.H.; Nan, Y. Improving the efficiency of enterprise technology application and strengthening the construction of human capital. *China Soc. Sci.* 2019, *6*, 63–84.
- 42. Perez-Sebastian, F. Public Support to innovation and imitation in a non-scale growth model. J. Econ. Dyn. Control. 2007, 31, 3791–3821. [CrossRef]
- 43. Peng, Y.Z.; Tao, C.Q. Can digital transformation promote enterprise performance?—From the perspective of public policy and innovation. *J. Innov. Knowl.* **2022**, *7*, 100198. [CrossRef]
- Mayo, J.W.; Wallsten, S. From network externalities to broadband growth externalities: A bridge not yet built. *Rev. Ind. Organ.* 2011, 38, 173–190. [CrossRef]
- 45. Zhao, T.; Zhang, Z.; Liang, S.K. Digital economy, entrepreneurship activity and high quality development—Empirical evidence from Chinese cities. *Manag. World* **2020**, *36*, 65–76.
- Zhu, Z.C.; Liu, B.; Yu, Z.X.; Cao, J.H. Effects of the Digital Economy on Carbon Emissions: Evidence from China. Int. J. Environ. Res. Public Health 2022, 19, 9450. [CrossRef]
- 47. Khan, A.; Wu, X.M. Digital economy and environmental sustainability: Do information communication and technology (ICT) and economic complexity matter? *Int. J. Environ. Res. Public Health* **2022**, *19*, 12301. [CrossRef]
- 48. Zhang, S.S.; Ma, X.Y.; Cui, Q. Assessing the impact of the digital economy on green total factor energy efficiency in the post-COVID-19 era. *Front. Energy Res.* 2021, *9*, 2296. [CrossRef]
- 49. Mubarak, M.F.; Tiwari, S.; Petraite, M.; Mubarik, M.; Rasi, R.Z.R.M. How Industry 4.0 technologies and open innovation can improve green innovation performance. *Manag. Environ. Qual.* **2021**, *32*, 1007–1022. [CrossRef]
- 50. Li, H.; Zou, Q. Research on environmental regulation, resource endowment and urban industrial transformation—Based on comparative analysis of resource-based cities and non resource-based cities. *Econ. Res.* **2018**, *11*, 182–198.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.